Author's response to reviews

Title: Attitudes to kidney donation among primary care patients in rural Crete, Greece

Authors:

Emmanouil K Symvoulakis (symvouman@yahoo.com)
Ioannis D Komninos (yiankom@yahoo.co.uk)
Nikos Antonakis (antonakisnikos@yahoo.gr)
Myfanwy Morgan (myfanwy.morgan@kcl.ac.uk)
Athanasios Alegakis (thalegak@med.uoc.gr)
Emmanouil Tsafantakis (etsafantakis@yahoo.gr)
Marios Chatziarsenis (chatz_mar1@yahoo.com)
Anastas Philalithis (tassos@med.uoc.gr)
Roger Jones (roger.jones@kcl.ac.uk)

Version: 4 Date: 28 January 2009

Author's response to reviews: see over
Date: 28 January 2009

BMC PUBLIC HEALTH

Manuscript No: 1108705289207643

Title: Attitudes to kidney donation among primary care patients in rural Crete, Greece

Author(s): Emmanouil K Symvoulakis, Ioannis D Komninos, Nikos Antonakis, Myfanwy Morgan, Athanasios Alegakis, Emmanouil Tsafantakis, Marios Chatziarsenis, Anastas Philalithis and Roger Jones

Dear editor

We would like to submit the corrected manuscript, entitled “Attitudes to kidney donation among primary care patients in rural Crete, Greece”. We added the tables at the end of the manuscript, after references section. The requested details were introduced in the affiliation 1. Affiliations 2 and 3 are referred to the primary care settings involved in the study (with the existing administrative links in the case of affiliation 2) and not to specific departments. Affiliation 2 was adjusted to the general style. The manuscript was typographically checked and very few minor language style corrections were performed.

We are looking forward to hearing from you.

Yours sincerely,

Emmanouil K. Symvoulakis